Author's response to reviews

Title: Double vs single internal thoracic artery harvesting in diabetic patients: role in perioperative infection rate

Authors:

Marco Agrifoglio (marco.agrifoglio@unimi.it)
Matteo Trezzi (mtrezzi@unimi.it)
Fabio Barili (fabarili@libero.it)
Luca Dainese (luca.dainese@ccfm.it)
Faisal H Cheema (fhc202@columbia.edu)
Veli K Topkara (vtk210@columbia.edu)
Chiara Ghislandi (cghisl@unimi.it)
Alessandro Parolari (aleparo@unimi.it)
Gianluca Polvani (glp@unimi.it)
Francesco Alamanni (fa@unimi.it)
Paolo Biglioli (pb@unimi.it)

Version: 3 Date: 6 May 2008

Author's response to reviews: see over
To the Editor

The Journal of Cardiothoracic Surgery

Milan, May 6th, 2008

Dear Sir,

We submit a revised version of our paper “Double vs single internal thoracic artery harvesting in diabetic patients: role in perioperative infection rate” for publication on The Journal of Cardiothoracic Surgery.

We edited the manuscript accordingly with the formatting changes requested.

We look forward to hear from You soon,

Best Regards,

Fabio Barili, MD
Dept. of Cardiovascular Surgery, University of Milan
Centro Cardiologico Monzino IRCCS, Milan, Italy
Via Parea, 4 20138 Milan, Italy
Email: fabarili@libero.it
Phone: +39 02 58002355 Fax: +39 02 5801194
All the below formatting changes were performed.

Please remove the text 'Title page' 'Title' and 'Authors' from the title page.

Please remove the full stop from the end of the title in the manuscript and the submission system.

Please remove all authors qualifications (e.g. MD, PhD) from the authors list.

Please include the country at the end of the affiliation details for affiliation 2.

Email address- please include email addresses for all the authors on the title page.

Please remove (Abstract word count: 250) from the abstract title.

Please remove Text from the top of page 3.

References - References should be cited in the text using consecutive numbers in square brackets.
Please do not use superscript numbers. Also, please place the reference before the full stop. For example `text [1].

Competing interests - If there are none to declare, please write `The authors declare that they have no competing interests’.

Please remove the repeated authors list from the beginning of the authors' contributions section.

If you have no acknowledgments or funding information to give, you may remove the Acknowledgments section.

Reference 2- has this article been published? If so please complete the citation details.

References - The reference list should contain all authors' names, if there are 30 or below. The term “et al.” should not be used. Please update references 8, 9, 11, 12, 20, 26.

Please use sentence case for the table headings, for example change TABLE 1 to Table 1

Every table should be cited in the text at least once. We could not find table 3 cited in the text.